Author’s response to reviews

Title: Implementation of outpatient schema therapy for borderline personality disorder: study design

Authors:

Marjon Nadort (m.nadort@ggzingeest.nl)
Arnoud Arntz (arnoud.arntz@maastrichtuniversity.nl)
Johannes H. Smit (jh.smit@ggzingeest.nl)
Josephine Giesen-Bloo (j.giesen@lavori.nl)
Merijn Eikelenboom (m.eikelenboom@ggzingeest.nl)
Philip Spinhoven (SPINHOVEN@FSW.leidenuniv.nl)
Thea van Asselt (thea.van.asselt@mumc.nl)
Michel Wensing (M.Wensing@iq.umcn.nl)
Richard van Dyck (R.vanDyck@ggzingeest.nl)

Version: 2 Date: 23 September 2009

Author’s response to reviews:

Changes of references 22, 31 and 40. Instead of submitted I mentioned the dissertation in which the papers were published in 2006.

Reference 19 and 54 still in press (no changes).

I uploaded the two figures as separate figure files.

I deleted the vertical text in the flow chart.

I changed references with et al. and listed all the authors.

I changed the title, I used sentence case, I removed all unnecessary capitals.

I removed the qualifications of the authors.

I removed the keywords.

I removed the figure an table placement in the text.

I changed the roman numeral of table 1.